



Article Strategic Behavior of Competitive Local Citizen Energy Communities in Liberalized Electricity Markets

Hugo Algarvio * , António Couto , Fernando Lopes 🕒 and Ana Estanqueiro D

LNEG-National Laboratory of Energy and Geology, Est. Paço Lumiar 22, 1649-038 Lisbon, Portugal

* Correspondence: hugo.algarvio@lneg.pt

Abstract: The liberalization of energy retail markets empowered consumers with the ability to be part of new emerging entities, such as Citizen Energy Communities. With the increasing penetration of decentralized variable generation, communities have the advantage of incentive local carbon neutrality and sustainability. Local generation reduces transport grid usage and costs to consumers. Furthermore, worldwide legislation incentives energy communities by providing them discounts to other fee parts of the tariff apart from wholesale prices. This paper presents a model of strategic behavior, investment, and trading of energy communities. The model comprises the investment in local renewable generation, the design of competitive tariffs, and strategic bidding on wholesale markets. Consumers have an optimization model that selects the retail tariff that minimizes their costs with energy. These models are tested using data from Portuguese consumers and the Iberian electricity market. Results from the study indicate that inflexible consumers may reduce their costs above 50% when using demand–response, adapting themselves to local production and wholesale prices.

Keywords: citizen energy communities; decentralized renewable generation; electricity markets; energy tariffs; local consumers; strategic bidding



Citation: Algarvio, H.; Couto, A.; Lopes, F.; Estanqueiro, A. Strategic Behavior of Competitive Local Citizen Energy Communities in Liberalized Electricity Markets. *Energies* **2024**, *17*, 1863. https:// doi.org/10.3390/en17081863

Academic Editor: Jay Zarnikau

Received: 23 February 2024 Revised: 28 March 2024 Accepted: 6 April 2024 Published: 13 April 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

The liberalization of retail and wholesale markets increases competition in electricity markets [1]. Suppliers, producers, aggregators, retailers, and big consumers trade energy in competitive wholesale markets [2]. Wholesale markets usually are divided into long-term derivative and bilateral markets, short-term marginal day-ahead and intraday markets, close to real-time marginal real-time markets and continuous pay-as-bid intraday markets, and real-time ancillary services [3,4]. The liberalization of retail markets increased the number of different suppliers, such as retailers, aggregators, traders, and, more recently, energy communities, also known as "Citizen Energy Communities" (CECs) or Local Energy Communities. Furthermore, retail competition increases the number of tariffs proposed to consumers in retail markets. Consumers may choose the best tariff according to their load profiles [5,6]. Suppliers shall study consumers' behavior and propose attractive tariffs to persuade them to join their portfolios [6–8]. The switching behavior of consumers increases retail competition and reduces tariffs [8,9].

The worldwide goals for a carbon-neutral society and the decreasing costs of new renewable plants, e.g., solar PV plants, increases the number of decentralized power plants and the possible competitiveness of CECs [10–12]. Furthermore, electrification and sector coupling are increasing the need for electrical energy for cooking, heating, cooling, electric vehicles, heat pumps, etc. [13]. Indeed, investments in variable renewable energy sources (vRES) and electrification reduce pollution, energy poverty, and outside dependency by reducing the need for fossil fuels [14–16].

Traditionally, energy resources were extracted or generated in large centralized industries. In the case of electrical energy, big power plants produced energy, which was transported and then distributed to end-use consumers [3]. Centralized and liberalized energy markets have been created to increase competition, but local markets and balance are important to avoid grid congestion and incentivize local sustainability [17–19]. The increasing number of decentralized energy resources increases the complexity of managing distribution grids [20]. Furthermore, it originates a new paradigm where energy flows go from distribution to transmission power grids. The increasing number of decentralized generation and demand led governments to provide economic incentives to local citizen energy communities (CECs) [21,22]. They have grid access discounts, incentivizing their local self-consumption and energy sustainability, which may reduce the usage of high-voltage power grids and avoid their consequent congestion [23,24]. Thus, flexible CECs are incentivized to help the management of the local grid, reducing the effort of local system operators [20,25].

Against this background, consumers are incentivized to be part of CECs and to manage their local resources [26–29]. Indeed, the European legislation's flexibility regarding CECs allows the formation of multiple types of energy communities, from a microgrid community with one owner similar to a prosumer to a national-level virtual community similar to a retailer [30–34]. The main issues about the community business models consist of their formation, weight, and scope [35–38]. Multiple CECs' business models have been proposed in the literature [19,39,40]. Small CECs do not have all economic benefits by needing intermediaries to participate in wholesale markets, while virtual national-level CECs may have grid access and usage restrictions and costs [32,41–44]. One of the main problems of local communities are their lack of expertise to become active market players [42,45]. Some CECs are formed by retailers, providing small discounts to consumers compared with their potential cost reduction [9]. The formation and management of CECs are important to guarantee their economic sustainability [35–38]. CEC's agents shall be equipped with decision protocols, intra-team strategies, forecast methodologies, and strategic bidding [42,46–48]. Furthermore, self-consumption CECs shall be capable to explore vRES generation complementarity, optimal investment models, and resource management [36,49,50]. Legally, consumers can choose their suppliers without a fixed loyalty period. Thus, CECs with self-consumption shall propose attractive, stable, and diversified tariffs to guarantee the loyalty and retention of their members [8,9].

This article focuses on the competitiveness of local CECs. In this case, the community is considered large enough to participate in wholesale markets and invest in local generation, benefiting from reduced grid usage and access costs and contributing to the local energy balance [51,52]. This type of community proposes tariffs similar to retailers, attracting consumers to a well-known market environment [6]. However, it also proposes tariffs indexed to local generation and dynamic market prices, preparing consumers for uncertain productions and prices of power systems with high vRES penetrations. Specifically, the purpose of this paper is three-fold:

- Design of a business model for agent-based CECs, comprising competitive tariffs, optimal investment in local vRES, and strategic bidding of local net load by adapting different forecast methodologies;
- 2. Design of a model simulating consumers' behavior when selecting tariffs and investing in self-consumption;
- 3. A study to test the previous models in the Iberian market of electricity (MIBEL) considering data of: (i) Portuguese consumers, (ii) market prices, and (iii) the vRES investment, operational, and maintenance costs.

The remainder of this paper is structured as follows. Section 2 presents a literature review regarding citizen energy communities. Section 3 presents the main concepts of electricity markets, bilateral contracts, and tariffs. Section 4 presents the optimal assets investments of CECs, tariffs they propose to members, and their strategic bidding. Section 5 presents the consumers' behavior on retail markets. Section 6 presents a study testing the negotiation models. Section 7 summarizes the main conclusions of the work.

2. Literature Review

The liberalization and decarbonization of the energy sector led to the establishment of new, decentralized, small market players, such as prosumers. Deng and Lv [14] presented a review of optimization models for power system planning with increasing penetrations of vRES. Hampton et al. [9] presented a review of the three main retail strategies to engage consumers: client focus, tariff designs, and innovation. Small players may not have the expertise and weight to participate in wholesale markets. To avoid the use of intermediaries, they may form coalitions, such as local CECs.

Consumers are being incentivized to form CECs, with CECs being well spread worldwide [19,34,39]. The European Union (EU) legislation strongly incentivizes the formation of CECs, from local microgrids to a national level (similar to retailers) [30,31]. Member states of the EU, such as Italy and Portugal, are investing billions of euros in the support of CECs and self-consumption. The Italian National Recovery and Resilience Plan allocated 2.2 billion EUR to incentive CECs [53]. In Portugal, the regulator provides 50% and 100% discount in the General Economic Interest Cost (GEIC) as part of the grid-access price for CECs and CECs with self-consumption, respectively. This cost is paid by end-use consumers due to over costs of the tariff deficit related to the renewable support schemes, capacity mechanisms, and other externalities [21]. In the United States, the Inflation Reduction Act of 2022 reserved 370 billion USD for financial incentives to support CECs that invest in renewable generation [54]. In Australia, the Alkimos Beach trial, allowed a community of 119 households to share a battery during five years, resulting in savings of 81 thousand AUD, and a significant decrease in their peak net load [55]. In India, the prosumers of a CEC in Lucknow with rooftop PV sell their excess energy between them for a price 43% lower than the wholesale price [56]. In Brazil, the RevoluSolar energy initiative promoted energy sustainability in a favela by supporting a local CEC of 30 families with access to solar PV generation [57]. In South Africa, a CEC composed of 30 consumers has been created with access to an embedded microgrid of rooftop PV aiming at provide local energy sustainability [58,59].

Ceglia et al. [38] identified that the main barriers for the dissemination of CECs are lack of information and technical skills after analyzing 111 references, since they are mainly composed of consumers and prosumers. Lazdins, Mutule, and Zalostiba [41] analyzed 68 references to identify the need for a common policy framework and peer-to-peer markets as the leading solutions for CECs dissemination. Zepter et al. [29] studied a set of residential prosumers in London, concluding that by investing in storage and considering peer-to-peer trades, each prosumer can reduce its costs with electricity by 30%. However, if they they act coordinated as a community, they can save around 60%.

Modeling the business models of CECs has been the focus of the most recent literature. Gjor et al. [23] presented a review of the actors and their roles and interactions in CECs. They benchmarked the literature concerning the objectives, models, and constraints of CECs designs, reviewing the numerical indicators that quantify the economic, environmental, social, and technical impacts of CECs.

Kubli and Puranik [34] identified 25 different business model designs after analyzing the implemented cases of 90 CECs. They concluded that generating renewable energy, increasing self-consumption, reducing energy costs, and coordinating partners are the main drivers for the current business models of CECs, mainly for residential prosumers. However, prominent business models such as providing system services, increasing grid reliability, and being active market players are also in the scope of some of the studied CECs.

Schwidtal et al. [19] presented 221 business options between 7 CECs' members or partners (consumers, prosumers, aggregators, retailers, the grid platform, and storage operators) after reviewing 135 journal papers. They concluded that the academic studies consider three market models: (i) peer-to-peer, (ii) community self-consumption, and (iii) transitive energy. Prosumers are considered the central players of these market models. While peer-to-peer models consider only local transactions of energy between local players, the other market models are more complex. Community self-consumption models consider

the coordination of local partners together with system operators to gather local flexibility and providing services to increase the grid reliability. Transitive energy models consider a significant diversification of business models, considering aggregators/CECs of local partners the key facilitators of their active participation in the provision of system services and participation in wholesale markets.

Reis et al. [39] identified 8 business models of 36 CECs, concluding that the majority focus on self-consumption and a few on important models to the energy transition, such as aggregation, energy efficiency, flexibility, and mobility.

Some works modeled the scope and operation of consumers and local CECs. Wei et al. [47] presented a review of 128 forecast models of demand and concluded that forecasts of small-scale consumers have higher errors than forecasts of large-scale consumers, indicating that the best forecasts have a mean absolute percentage error (MAPE) until 10%. Hong et al. [8] presented a bilevel game-theoretic model for strategic retailers participating in wholesale and local electricity markets. They concluded that modeling the consumers' behavior when choosing tariffs is very important to increase retail competition and decrease retail prices.

Most CECs consider small communities of consumers and prosumers to consume local renewable generation and reduce costs. These CECs may need a profit-seeking intermediary to trade their excess or lack of energy. The model presented in this work considers an active agent-based CEC that can cooperatively invest and manage local renewable generation and participate in wholesale markets to minimize costs. Against this background, modeling communities and their dynamic distributed players and resources as software agents, and equipping them with the required trading models can be relevant for their efficient participation in competitive markets. Klusch and Gerber [28] used the concept of dynamic coalition formation to simulate the alliance of partners as software agents who want to achieve a common goal by cooperatively responding to events. Sanchez-Anguix et al. [48] designed a negotiation model of alliances that uses bottom-up methodologies applied to achieve Pareto-optimal agreements. Mansour et al. [60] presented a hybrid negotiation strategy to define the offers and counter-offers of software agents. The model consists of preference-based and fuzzy-similarity methods used for quantitative and qualitative issues, respectively. The authors concluded that the hybrid strategy improved the negotiation output of agents when compared to other strategies.

This contribution extends prior research of the main authors of this work within the scope of CECs, namely:

- Their regulatory framework, indicating their role in power systems and their potential members, benefits, and functions [21].
- A review and model for their agent-based formation and management of their resources [27].
- Their agent-based model equipped with coalition formation and management models, decision protocols, and intra-team strategies [36].
- Their strategic bidding process, considering their active participation in wholesale markets without the need for intermediaries [46].
- Their interaction and decision protocols in an automated negotiation between alliances and opponents. It presented a decision-support model for energy alliances considering automated negotiation, enabling small market players to form and manage teams and their decisions [42].

These previous studies identified the economic benefits for consumers being part of active CECs by allowing their competitive participation in wholesale markets. The work developed in these studies together with the developed modeling frameworks of retail competition [6], contract characteristics [61], and wholesale markets [62] are used in the work presented in this paper.

3. Electricity Markets, Bilateral Contracts, and Tariffs

Consumers sign private bilateral contracts with suppliers, being subject to energy tariffs. Those tariffs may have different risk levels for consumers [5]. They have no risk

if fixed and high risk if indexed to wholesale prices. Suppliers define tariffs according to the expected prices of wholesale markets plus a risk premium to compensate for their risk exposure [61,63]. Furthermore, the energy part of tariffs is subject to more fees than wholesale energy prices, such as grid access, which incentives consumers to invest in self-consumption.

3.1. Electricity Markets and Bilateral Contracts

Electricity markets are divided into spot, continuous, balancing, derivatives, and bilateral contracts markets [2,4].

Spot markets are auction-based marginal markets such as day-ahead, intraday, and realtime markets. Continuous markets consider the direct match of opposite bids using the pay-as-bid scheme. Balancing markets balance demand and supply during real-time operation, guaranteeing frequency stability [62,64]. Derivatives and bilateral contracts markets consider the negotiation of standard and non-standard bilateral contracts, respectively [61].

Bilateral contracts can be standard, non-standard, physical, or financial [3]. Standard bilateral contracts can consider physical and financial trading of forwards, futures, options, and swaps (also known as contracts for differences). Using forward contracts, traders define the price at present to a future period. In futures contracts, traders define a price subject to financial settlement in the future. Options contracts are divided into: (i) put and (ii) call options, i.e., option to sell or buy an asset, respectively. The buyers of those options guarantee a fixed future asset price by paying a premium, reserving the choice to use the option or not. Sellers receive a premium but are obliged to follow the decision from buyers. Swaps are only subject to a financial settlement according to the strike price. Sellers compensate buyers when spot prices are higher than strike prices. Otherwise, buyers compensate sellers [61]. Physical contracts consider energy delivery and financial settlement, while financial contracts only consider the latter. They are used to hedge against the volatility of spot markets. Non-standard contracts, such as power purchase agreements, can be traded both on wholesale and retail markets.

Consumers negotiate private bilateral contracts with retailers and other intermediaries to hedge against the volatility of wholesale prices. These private contracts are non-standard contracts, which means that several terms and conditions can be agreed (see, e.g., [61]). Typically, conditions such as the type of tariff are fixed, but prices may suffer periodic updates (e.g., yearly updates).

3.2. Tariffs

Tariffs are divided into two parts, a fixed price for contracted power, and a variable price per unit of consumed electricity [5]. The variable price of the tariff is divided into two parts, the energy part from wholesale market prices and the other part from the system costs. The other part of the variable term is fixed and defined by the regulator. It includes grid access, considering the GEIC, commercialization and the global system, transportation, and distribution grid usage [65]. Thus, grid investment, maintenance and operation, and renewable support are included in this fixed part of the variable term.

The most used tariffs are single and time-of-use (TOU) tariffs. Typically, TOU tariffs consider two-rate, three-rate, and four-rate tariffs (involving peak, mid-peak, off-peak, and super off-peak periods). Consumers select TOU tariffs when they consume more during off-peak periods or have the flexibility to do so. These tariffs are used as a demand-side response (DSR) strategy, when regulators and retailers try to adapt typical consumption behaviors to reduce the need to use traditional generation. Typically, this kind of long-term demand response (DR) induces a change in the consumption profiles of consumers according to each net load period. Net load is the difference between consumption and vRES generation. High/low net loads mean energy scarcity/excess of vRES generation, respectively. Energy scarcity/excess induces high/low wholesale prices and the need for a high/low production of traditional generation, respectively. In power systems with relevant penetrations of variable renewable energy, wholesale prices are more volatile. Thus, the use of TOU tariffs may not reflect the real-time power system needs, by not

providing the right price signals to consumers. In these power system, retailers and regulators may propose tariffs with precise short-term price signals to hedge against the volatility of wholesale prices.

On liberalized markets, retailers are also proposing tariffs indexed to wholesale market prices, such as real-time pricing (RTP) and day-ahead hourly (DAHP) and monthly pricing (DAMP) tariffs. These tariffs reduce retailers' risk and incentive flexible consumers to change their short-term consumption behavior (short-term DR). These tariffs also include the commissions from retailers. However, those commissions are fixed and lower when compared with single and TOU tariffs, subject to high-risk premiums [5,21,66].

In these indexed tariffs, consumers do not know precisely the price to pay before consuming because they are subject to energy losses computed by the regulator. DAHP tariffs are the ones that provide consumers more precise information about hourly prices between 12 and 36 h before real-time dispatch. They consider the day-ahead hourly price plus a fixed value for real-time imbalances. These tariffs allow consumers to schedule their consumption. However, they do not provide the most precise price signals, since forecast errors can lead to periods of scarcity/excess of energy according to the programmed dispatch, which may lead to higher/smaller imbalance costs. RTP tariffs are the ones that provide the most accurate prices according to the actual wholesale price of energy. However, retailers can only compute those prices after consumption, which does not give consumers the best information to schedule consumption. These tariffs are adequate for flexible consumers that can comply with their programmed dispatch without deviations. These consumers may avoid the payment of risk premiums to protect themselves from paying imbalance penalties by choosing RTP tariffs. DAMP tariffs consider the average monthly day-ahead prices of the entire portfolio plus a fixed fee for real-time imbalances, which do not incentive consumers to change their consumption behavior.

Notice that all consumers have indexed tariffs available. Usually, the only risk-free tariffs available for medium voltage consumers are four-rate TOU tariffs. Consumers connected to the low voltage grid may only choose from single to three-rate TOU tariffs and high-voltage consumers may choose hourly tariffs. Table 1 presents the main characteristics of each type of tariff considering the risk and price signals given to each party after analyzing the literature [1,5–9].

Tariffs	Buyer's Risk	Seller's Risk	Price Signals	Incentive Long-Term DR	Incentive Short-Term DR	Incentive Real-Time DR
Single	No	High	No	No	No	No
TOU	No	High	Low	High	Low	No
DAHP	Medium	Low	High	No	High	No
DAMP	High	Low	No	No	No	No
RTP	High	No	Medium	No	Medium	High

Table 1. Main characteristics of the type of tariffs.

The requested risk premiums by suppliers depend on their exposure to risk. Thus, single and TOU tariffs may be higher than other types of tariffs during stable periods. However, they protect consumers in the case of unexpected price spikes. Actual suppliers may protect consumers by offering fixed tariffs for a period (e.g., one year) but do not incentive consumers to update their tariffs.

Power systems with high penetrations of vRES need supply and demand flexible resources. Nowadays, investing in storage assets such as batteries is expensive. Thus, suppliers and operators shall incentivize flexible consumers to accept indexed tariffs and demand response programs to avoid price spikes and load shedding in scarcity periods.

The tariff structure incentivizes prosumers and CECs to invest in storage to avoid selling their excess energy at low prices and buying their missing energy at high prices [29]. However, regulators shall consider dynamic tariffs to incentivize consumers to use their

flexible energy resources to help the management of the local grid. Furthermore, distribution system operators shall incentivize consumers to participate in local system services. Thus, tariffs shall provide dynamic price signals according to the real-time power system needs to avoid increased energy costs because of investments in centralized generation backup and storage assets.

4. Optimal Assets Investments, Tariffs, and Strategic Bidding of Communities

Being part of Citizen Energy Communities is one of the most competitive solutions for consumers [21,22,42]. Besides reducing the price of the other fees of the energy part of the tariff, CECs may invest in local generation, proposing more competitive green tariffs to local consumers. Consuming locally in self-consumption CECs means that consumers may only pay local grid usage besides the investment costs in local generation and storage assets. Furthermore, CECs may invest in a cheaper and more efficient large-scale generation when compared to the small-scale units available for consumers. Even considering these investments, CECs shall propose competitive, stable, and diversified tariffs to be more attractive to local consumers than regulated and retail tariffs. Against this background, CECs may investigate potential investments in local assets to feed their portfolio of consumers when compared to wholesale prices.

4.1. Optimal Asset Investments

The potential investment of CECs in local assets affects the proposed tariffs. To be attractive and stable, tariffs proposed by CECs may keep the best features of actual tariffs and improve their worst features. Thus, they shall be fixed for a period to protect consumers in case of price increases but also shall recommend consumers to update tariffs in case of price decreases.

One way to propose more competitive and stable tariffs to local consumers consists of satisfying their consumption using local variable renewable energy. vRES have substantial fixed costs but near-zero marginal costs, which means that investors know exactly their annual costs with these technologies. Their main source of uncertainty is their hourly prediction, being their annual production stable. However, it is possible to compute their levelized costs of energy (LCOE) and compare them with wholesale prices to evaluate a potential investment. The LCOE considers that investors recover all annual, *n*, fixed, cf_n^v , and variable, cv_n^v , costs of a technology, *v*, during its life cycle, *N*, according to its production, q_n^v , and discount rate, *r*:

$$LCOE = \frac{\sum_{n=1}^{N} cf_{n}^{v} + cv_{n}^{v} \times (1+r)^{-n}}{\sum_{n=1}^{N} q_{n}^{v} \times (1+r)^{-n}}$$
(1)

The yearly production, q_n^v , is computed considering the hourly generation, q_t^v , of the technology as follows, with *T* being the number of hours of that year:

$$q_n^v = \sum_{t=1}^T q_t^v \tag{2}$$

For investment purposes, LCOE can be predicted using the expected capacity factor of the technology, \hat{Pf}^{v} , its nominal power Pn^{v} , and the number of hours per investment period:

$$\hat{q}_n^v = \hat{Pf}^v \times Pn^v \times T \tag{3}$$

The optimal investment solution uses historical generation data of each technology, v, using per unit production, $\hat{q}_{t,pu}^v$, to extrapolate and forecast its future production, \hat{q}_t^v . Furthermore, it uses forecast of long-term portfolios consumption, \hat{q}_t , and levelized wholesale market prices, \hat{p}_t^s , from each spot market, $s \in S$, using multivariate time series presented in previous works, to compute the optimal capacity of each technology, Pn^{v} , as follows [5,6]:

$$\min_{Pn^{v}} c = \sum_{t=1}^{T} \hat{p}_{t}^{S} \times (\hat{q}_{t} - \sum_{v=1}^{V} \hat{q}_{t}^{v}) + \sum_{v=1}^{V} \hat{q}_{t}^{v} \times LCOE^{v}$$
(4)

$$\hat{q}_t^v = P n^v \times \hat{q}_{t,pu}^v \tag{5}$$

$$\hat{p}_t^S = \frac{\sum_{s=1}^S \hat{p}_t^s \times \hat{q}_t^s}{\hat{q}_t} \tag{6}$$

subject to:

$$\sum_{i=1}^{T} \hat{q}_t \ge \sum_{t=1}^{T} \sum_{v=1}^{V} \hat{q}_t^v \tag{7}$$

where *c* is the expected total cost of the community with the energy required by its consumers and the investment in vRES and \hat{q}_t^s is the expected quantity transacted on each spot market.

If the LCOE of technologies is lower than wholesale prices, the optimization may suggest investment in high vRES' capacities than needed to sell energy excess. Equation (7) constrains the limited investment to the expected quantity required by the consumers of the community.

4.2. Community Tariffs

To attract local consumers to being part of the community has been proposed single, four-rate TOU, DAHP, and RTP tariffs. These tariffs are valid for a specific period, being periodically updated. Members may change their tariff on all updates or be forced to accept new prices after their tariffs' valid period. Single and TOU tariffs are associated with vRES investments, while DAHP and RTP tariffs are indexed to markets. Single and TOU tariffs consider each vRES technology or a hybrid solution that integrates all vRES.

Suppliers are exposed to risks, such as wholesale price volatility and consumption uncertainty in fixed retail tariffs. Furthermore, if CECs invest in local generation, they are also exposed to uncertain local production. By consuming locally, members have discounts related to the other fees part, p_T^{CEC} , of the energy tariff, p_T^e , by only using the local grid with a limited number voltage levels, *U*. The other fees part of the energy tariff, p_T^f , is computed according to the expected local net load of the community.

It is important to notice that the other fees component of the energy tariff to propose in the fixed single and TOU tariffs are based on forecast data (conditions 1–3, but check the following plan template presented in Equation (8), while in the case of indexed tariffs observed values are used (condition 4). Forecast data of market prices is also used, as well as local demand and vRES generation to compute the expected costs with the other fees part of the tariff (conditions 1–3). Thus, if the expected generated energy is higher than the expected energy consumed by members, they will only pay the usage of power grids to transact that energy (condition 1). If the expected generated energy is zero, they will pay the fixed CEC fee, p_T^{CEC} , for a unit of consumed energy. Otherwise, members with a tariff associated with a local vRES technology have priority in consuming its energy (condition 2). Furthermore, members with tariffs associated with hybrid technologies (condition 3) have priority regarding consumers with indexed tariffs (condition 4). If the expected energy produced by local vRES are not enough to satisfy the previous groups of consumers, it is applied pro rata, i.e., generation is distributed uniformly per each priority group using the weight, w. The previous conditions allow to compute the other fees part of the tariff according to the following plan template (PT):

$$PT: p_{T}^{f} = \begin{cases} 1. \ \sum_{u=1}^{U} p_{T}^{u} \ \text{if} \ \sum_{v=1}^{V} \hat{q}_{t}^{v} \ge \hat{q}_{t} \\ 2. \ w \times \sum_{u=1}^{U} p_{T}^{u} + (w-1) \times p_{T}^{CEC} \ \forall \hat{q}_{t}^{T_{v}} \ \text{and} \ \sum_{u=1}^{U} p_{T}^{u} + p_{T}^{CEC} \ \forall \hat{q}_{t} \notin \{\hat{q}_{t}^{T_{v}}\} \ \text{if} \ \sum_{v=1}^{V} \hat{q}_{t}^{v} < \hat{q}_{t}^{T_{v}} \ \text{for} \ w = \frac{\sum_{v=1}^{V} \hat{q}_{t}^{v}}{\hat{q}_{t}^{T_{v}}} \\ 3. \ w \times \sum_{u=1}^{U} p_{T}^{u} + (w-1) \times p_{T}^{CEC} \ \forall \hat{q}_{t}^{T_{h}} \ \text{and} \ \sum_{u=1}^{U} p_{T}^{u} + p_{T}^{CEC} \ \forall \hat{q}_{t} \notin \{\hat{q}_{t}^{T_{v}}, \hat{q}_{t}^{T_{h}}\} \ \text{if} \ \sum_{v=1}^{V} \hat{q}_{t}^{v} - \hat{q}_{t}^{T_{v}} < \hat{q}_{t}^{T_{h}} \ \text{for} \ w = \frac{\sum_{v=1}^{V} \hat{q}_{t}^{v} - \hat{q}_{t}^{T_{v}}}{\hat{q}_{t}^{T_{h}}} \end{cases}$$

$$(8)$$

$$4. \ w \times \sum_{u=1}^{U} p_{T}^{u} + (w-1) \times p_{T}^{CEC} \ \forall q_{t} \notin \{q_{t}^{T_{v}}, q_{t}^{T_{h}}\} \ \text{if} \ \sum_{v=1}^{V} q_{t}^{v} < q_{t} \ \text{for} \ w = \frac{\sum_{v=1}^{V} \hat{q}_{t}^{v}}{\hat{q}_{t}} \end{cases}$$

where p_T^u is the fee for using the power grid with voltage u in period T, $\hat{q}_t^{T_v}$ and $\hat{q}_t^{T_h}$ are the forecast quantity consumed by the group of consumers with a tariff associated with technology v and hybrid h, respectively, and w is the relative net load, i.e., the weight of the expected local generation concerning consumption. Single and TOU energy parts of the tariff, p_T^e , are computed according to the expected vRES/Hybrid production, portfolio consumption, and wholesale prices plus a premium, k, to cover the risk exposition of the community, as follows:

$$[p_T^e] = \sum_{t=1}^T (\hat{p}_t^s + k) \frac{(\hat{q}_t - \sum_{v=1}^V \hat{q}_t^v) - \hat{q}_t^v \times LCOE^v}{\hat{q}_t}$$
(9)

In the case of single tariffs considering one or multiple technologies, v is fixed or $v \in \{1, V\}$, respectively. In the case of four-rate TOU tariffs, only the tariff value for each period is considered, obtaining a tariff matrix with four prices.

This work considers that CECs may also propose DAHP and RTP indexed tariffs. DAMP tariffs are neglected because they do not incentive local communities' cooperative behavior neither incentive consumers to respond to local energy scarcity/excess and grid congestion, and high/low wholesale prices. Because of potential penalties due to imbalances, CECs consider a risk premium, *k*, when computing the energy price of a DAHP tariff as follows:

$$p_t^e = p_t^{DAM} + k \tag{10}$$

where p_t^{DAM} is the hourly day-ahead energy price. RTP tariffs also consider dynamic ancillary services prices, p_t^{AS} , being the energy price consumers pay, computed as follows:

$$p_t^e = p_t^{DAM} + p_t^{AS} \tag{11}$$

Besides these prices, CECs also have to lead with energy losses. If energy is consumed and generated locally, losses are lower. Thus, it also incentives self-consumption. The strategic bidding of CECs to feed their members is presented in the next section.

4.3. Strategic Bidding in Wholesale Markets

CECs can be composed of consumers, prosumers, and distributed generation and storage. They can invest in self-consumption. By investing in self-consumption CECs know their annual financial obligations beforehand, reducing their exposure to the volatility of spot prices. However, CECs shall forecast their net load needs according to their consumption and production uncertainty. They may trade energy on marginal wholesale markets, such as the day-ahead (DAM) and intraday (IDM) spot markets. Furthermore, as balance responsible parties, to avoid the payment of penalties, they can use the intraday pay-as-bid continuous market to trade their expected deviations 1 hour ahead of real-time operation [62,67].

4.3.1. Clustering and Long-Term Forecast

The Multivariate Time series (MTS) forecast methodology has been adapted from [5] and [6] to predict the electricity cost and consumption per consumer, respectively. The presented data consider the gross consumption of different consumers. To verify the type of consumer, clustering methodologies are used to compare typical consumption profiles

with observed consumption. To do so, the Calinski–Harabasz (CH) index is used to obtain the optimal number of clusters, dividing each consumer by its consumption segment.

4.3.2. Strategic Bidding and Short-Term Forecast

This section presents a process for strategic bidding in wholesale electricity markets adapted from [46], considering that CECs can also invest in self-consumption (see also Figure 1). The process uses different types of data. It uses historical data to forecast the next-day consumption in the DAM based on a forced forecast. It selects from the database the most recent hour with an hourly consumption, *h*, according to the type of forecast day (D): weekday (W), Saturday (S), Sunday (U), or holiday (H). Considering the database with the historical daily consumption data, $D = \{W, S, U, H\}$, the formulation to obtain the forecast is:

$$\hat{q}_t = q_{t-h}, \forall h \in \mathcal{D} \tag{12}$$

subject to:

$$\min_{q_{t-h}} h, \{h | (\hat{q}_t \land q_{t-h}) \in (\mathcal{W} \lor \mathcal{S} \lor \mathcal{U} \lor \mathcal{H})\}$$
(13)



Figure 1. Strategic bidding process of CECs in wholesale markets.

For every time period, CECs can have multiple generation technologies v. Thus, the total quantity of electricity expected to be produced by all technologies, \hat{q}_t^V , is used to compute the bids to each period of the DAM, q_t^{DAM} . The wind and solar PV forecasting methodology employed a sequential forward feature selection (SFFS) algorithm to select meteorological parameters from a numerical weather prediction model, as described in detail in [68]. The goal was to optimize a predefined objective function, specifically to minimize the root mean square error (RMSE). The forecasting process began by subjecting each meteorological parameter to Principal Components Analysis (PCA). The PCA identified principal components based on their correlations, which were then ranked. These ranked principal components were subsequently used as inputs for the SFFS algorithm, which employed an artificial neural network statistical approach to generate the forecasts. A detailed description of the methodology is provided in [68]. Here, the methodology was also adapted to the solar power forecast case by including additional meteorological parameters such as cloud cover at different heights, allowing to compute the expected generation of solar PV and onshore wind technologies:

$$q_t^{DAM} = \hat{q}_t - \hat{q}_t^V \tag{14}$$

$$\hat{q}_t^V = \sum_{v=1}^V \hat{q}_t^v \tag{15}$$

For each continuous intraday period, t, the forecast, q_t^{cont} , uses the most updated local consumption and generation information to forecast the net load of the CEC and submit bids for the required electricity session. The intraday methodology has been adapted from a forecast methodology for retailers [6]. CECs compute the bids to submit to the continuous intraday market considering the historical net load behavior, y_t . The energy price, p_t^{cont} , is assumed to be the day-ahead price, q_t^{DAM} . The energy quantity, q_t^{cont} , is computed as follows:

$$q_t^{cont} = q_{t-1} - \sum_{v=1}^{V} q_{t-1}^v \times y_t - q_t^{DAM} - \sum_{v=1}^{V} \hat{q}_t^v$$
(16)

$$y_t = \frac{\bar{q}_t - \bar{q}_t^V}{\bar{q}_{t-1} - \bar{q}_{t-1}^V}$$
(17)

where \bar{q}_t and \bar{q}_t^V are the average members' consumption and generation on time *t*, respectively. Then, the real-time imbalance, q_t^{dev} , of period *t*, is computed considering the difference between the real-time consumption of the CEC, q_t , and the final programmed dispatch, q_t^{prog} , respectively:

$$q_t^{dev} = q_t^V - q_t + q_t^{DAM} + q_t^{cont} = q_t^V - q_t + q_t^{prog}$$
(18)

Each time period balance responsibility of the CEC, C_t^{dev} , considering its deviations, q_t^{dev} , and the prices of the excess or lack of electricity, in cases of up, P_t^{up} , or down, P_t^{down} deviations, respectively, are computed as follows:

$$\begin{cases}
C_t^{dev} = q_t^{dev} P_t^{up}, & \text{for } q_t^{dev} > 0 \\
C_t^{dev} = \left| q_t^{dev} \right| P_t^{down}, & \text{for } q_t^{dev} < 0
\end{cases}$$
(19)

Each technology v has its own $LCOE^{v}$, and thus, each time period cost, C_t , of the CEC is:

$$C_t = \sum_{v=1}^{V} LCOE^v q_t^v + p_t^{DAM} q_t^{DAM} + p_t^{cont} q_t^{cont} - C_t^{dev}$$
(20)

To evaluate the performance of the forecast techniques are used two different indicators, MAPE and normalized RMSE (NRMSE) [6]:

$$MAPE = \frac{100\%}{T} \sum_{t=1}^{T} \left| \frac{q_t - \hat{q}_t}{q_t} \right|$$
(21)

$$NRMSE = 100\% \frac{\sqrt{\frac{1}{T} \sum_{t=1}^{T} (\hat{q}_t - q_t)^2}}{q_{max}}$$
(22)

where q_{max} is the maximum observed quantity. The value of \hat{q}_t depends on the time horizon of each market forecasts, being equal to q_t^{DAM} in the case of day-ahead forecasts and equal to q_t^{prog} in the case of continuous intraday forecasts.

Consumers may choose among multiple types of tariffs from communities and suppliers. Thus, the next section studies how consumers select their tariffs.

5. Consumers Costs and Behavior on Retail Markets

Usually, consumers select the most affordable tariff adapted to their consumption behavior. However, price-based DSR mechanisms involve suppliers proposing tariffs to change typical consumption behaviors. Consumers may change their consumption behavior if that results in substantial savings. Normally, energy prices can be dynamic, but grid access fees are defined by regulators. According, the consumer's cost, c, in period T, is computed as follows:

$$c_T = \sum_{t=1}^T T_t^e \times q_t \tag{23}$$

$$T_t^e = p_T^e + p_T^f \tag{24}$$

where T_t^e is the energy part of the tariff, divided into the energy price and other fees parts, p_T^e and p_T^f , per time *t*, respectively, and q_t is the quantity consumed.

In case of single and TOU tariffs, the energy and other fees prices are fixed for all periods. In case of indexed tariffs, only the other fees are fixed, with the energy price of a DAHP computed as follows:

$$p_t^e = (p_t^{DA} + p^{AS} + k) \times (1 + q_t^L)$$
(25)

where *k* is the supplier's risk premium (fixed margin) and q_t^L are the hourly losses defined by the regulator. This tariff incentives consumers to respond to dynamic day-ahead prices, being all variables except hourly losses known from 12 to 36 h before they consume. RTP tariffs are similar, with the difference that the prices of ancillary services are also dynamic, with the energy price consumers pay computed as follows:

$$p_t^e = (p_t^{DA} + p_t^{AS} + k) \times (1 + q_t^L)$$
(26)

This tariff increases the risk to consumers, since they only know the day-ahead price before consuming. Thus, it does not incentive consumers to respond to prices so much as DAHP, but may reduce energy costs by reducing the risk premium associated with fixed ancillary services prices. Suppliers submit bids for entire portfolios based on demand forecasts, not informing clients about their programmed dispatch. Hence, the costs of ancillary services are paid by all consumers, even if they have the flexibility for selfregulation. Thus, both CECs and flexible consumers benefit from potential cooperation.

The energy price of DAMP is computed considering the average monthly day-ahead price of the supplier's portfolio, \bar{p}_M^{DA} , as follows:

$$p_t^e = (\bar{p}_M^{DA} + p^{AS} + k) \times (1 + q_t^L)$$
(27)

This type of tariff does not incentive consumers to adapt their consumption because they are not aware of energy prices before consuming.

With the liberalization of the retail market, its increasing competition allows consumers to select among a significant number of tariffs. Allowing consumers to change between tariffs and retailers without constraints increase competition but also the risk to suppliers. Against this background, this work modeled the consumers' selection of the best tariff, $o \in O$, with O being all available options, as follows:

$$c_o = \sum_{t=1}^T T_{t,o}^e \times q_t \forall o \in O$$
(28)

The previous formulation computes the costs consumers will have or expect to have with all available tariffs, according to if they are using scheduled or expected consumptions, respectively. If they use scheduled consumption, they can be trustful about changing tariffs. Otherwise, they shall be careful and only change tariffs if expecting significant savings. Changing tariffs when expecting small savings may lead to extra costs if consumers consider expected consumptions. Accordingly, consumers may change to the potential best tariff, T_{T+1}^e , that minimizes their costs, as follows:

$$T_{T+1}^e = \min_{T_{T+1,o}^e} c_o$$
(29)

If consumers are part of an energy community, they may significantly reduce energy costs with other fees. Furthermore, if the community has self-consumption, consumers can practically eliminate those costs. Investing in self-consumption is another way consumers have to reduce/eliminate those costs. A similar optimal investment can be used to consumers regarding the one used by CECs in Equation (4), as follows:

$$\min_{Pn^{v}} c_{o} = \sum_{v=1}^{V} \hat{q}_{t}^{v} \times LCOE^{v} + \begin{cases} \sum_{t=1}^{T} T_{t,o}^{e} \times (\hat{q}_{t} - \sum_{v=1}^{V} \hat{q}_{t}^{v}) \text{ if } \sum_{v=1}^{V} \hat{q}_{t}^{v} < \hat{q}_{t} \\ \sum_{t=1}^{T} p_{t}^{DA} \times (1 - k_{t}^{q^{L}}) \times (\hat{q}_{t} - \sum_{v=1}^{V} \hat{q}_{t}^{v}) \text{ if } \sum_{v=1}^{V} \hat{q}_{t}^{v} \ge \hat{q}_{t} \end{cases}$$
(30)

subject to:

$$\sum_{v=1}^{V} \hat{q}_{t}^{v} \ge \sum_{v=1}^{V} \hat{q}_{T-1}^{v}$$
(31)

where $k_t^{q^L}$ is the defined risk premium if the consumer sells its excess energy. Normally, this value is fixed and defined by the regulator to compensate the losses and deviation costs originated by this energy. However, in CECs this energy can be traded locally. Thus, CECs may compute a dynamic periodic risk premium, which can be high in case of local energy excess and negative in case of local energy deficit, which incentive members with self-consumption to adapt their consumption to local needs. Equations (7) and (31) constrain the limited investment to the expected quantity and previous investment of each consumer, respectively.

This work presents two different optimization models to consumers because investing in self-consumption shall be optional (see Equations (29) and (30)). Even if investing in self-consumption is cheaper than all tariffs, consumers may not have the required budget.

6. Case Study

The study considers one Citizen Energy Community attracting 312 local consumers to its portfolio during 2019 (see [69] for more details about the consumers). It competes with the regulator and retailers' tariffs. These consumers are connected to the medium voltage 30 kV grid, meaning that some of them are part of aggregations of residential and small commercial consumers. Using the K-means clustering algorithm and the CH criterion, it is possible to divide them into the following five consumption segments: 10 industrial, 11 large commercial, 189 aggregations. Real consumption data have been used from 1 January 2012 to 31 December 2013 extrapolated to 2019 using MTS forecasts adapted from [5,6]. Furthermore, the market prices of MIBEL have been used, as well as the most competitive tariffs proposed by the Portuguese regulator and retailers (see the Data Availability Statement for details).

To compute the optimal investment assets of both CECs and consumers, we considered a discount rate of 4%, a life cycle of 25 years, and an investment payment of 20 years for all technologies. Furthermore, we considered the typical capacity factors and the 2019 fixed and operation and maintenance (O&M) marginal costs of the studied technologies in Portugal, as presented in Table 2 [70].

Accordingly, in the first step, the CEC computes the tariffs to propose to consumers, presented in the next section.

Technologies	Fixed Cost (€/kW)	O&M Costs (EUR/MWh)	Capacity Factor (%)	LCOE ₂₀₁₉ (EUR/MWh)	Voltage Level (kV)
Onshore Wind	1254	25.4	25	52.35	60
Large-scale PV	910	20	22	43.11	30
Rooftop PV	1220	20	18	66.33	-

Table 2. Main characteristics of the type of technologies.

6.1. Tariff Proposals

In Portugal, demand-side players have to pay for grid access and usage independently of being active market players, but legislation strongly incentive self-consumption. The regulator defined that consumers with self-consumption pays a risk premium of 10% for the excess of energy. In the case of communities without and with self-consumption the legislation provides them a 50% and 100% discount in the GEIC part of the variable cost of energy, respectively. In Portugal, the 2019 prices of each part of the variable term of energy for medium voltage consumers are equal to (all in EUR/MWh) [65]:

- Energy (only for regulated tariffs):
 - Peak: 77.2
 - Mid-Peak: 72.4
 - Off-Peak: 59.0
 - Super off-Peak: 54.7
- Grid access: 19.15
 - CEC: -5.36
 - CEC w/ self-consumption: -10.72
- Global system use: 6.37
- Transport grid use: 1.30
- Distribution grid use:
 - High voltage 60 kV: 0.60
 - Medium voltage 30 kV: 1.38

Consumers with self-consumption have the potential to reduce costs with energy, with the considered periods to apply four-rate TOU tariffs presented in Table 3.

Table 3. Electricity price and pricing periods defined by the regulator for 2019.

	Energy Price [EUR/MWh]	Working Day	Saturday	Sunday
Peak	106.15	17.00-22.00		
Mid-Peak	101.35	00.00-00.30 07.30-17.00 22.00-24.00	10.30–12.30 17.30–22.30	
Off-Peak	87.95	00.30–02.00 06.00–07.30	00.00-03.00 07.00-10.30 12.30-17.30 22.30-24.00	00.00–04.00 08.00–24.00
Super Off-Peak	83.65	02.00-06.00	03.00-07.00	04.00-08.00

The CEC computes annual tariffs according to a forecast of the market prices and local consumption using the presented MTS forecast methodologies. Furthermore, it forecasts the market values of each vRES technology according to its historical production. Single and four-rate TOU tariffs are subject to risk premiums of 6% and 5% over the expected energy costs of the variable term of the tariff, respectively. These risk premiums compensate for CEC's risk in forecasting energy prices, vRES generation, local consumption, and balancing

costs. The DAHP tariff is also subject to a risk premium of 5% over expected energy prices, while the RTP tariff does not have a risk premium.

Using the optimization model presented in Section 4.1, it is possible to compute the single and four-rate TOU (PV, Wind, and Hybrid) tariffs allocated to each vRES technologies to invest for. Hybrid tariffs have an optimal proportion of 77.6% PV and 22.4% onshore wind. We have used the per unit expected production values of each technology to plan its investment and compute the tariffs associated with local vRES production. The final investment in local vRES capacity depends if local consumers choose these tariffs. These tariffs compete with the regulator and the three most competitive retail tariffs. The regulator proposes a four-rate TOU tariff and retailers DAHP, DAMP, and RTP tariffs. The main characteristics of these tariffs are presented in Table 4.

Agents	Tariffs	Fixed Energy Price (EUR/MWh)	Indexed Energy Price (EUR/MWh)	Risk Premium and Commission
Regulator	TOU	$\{106.15; 101.35; 87.95; 83.65\}$	No	No
Retailers	DAHP DAMP	Other fees Other fees	Hourly day-ahead Monthly day-ahead	14 EUR/MWh 16.5 EUR/MWh
	RTP	Other fees	Hourly day-ahead and balancing	20 EUR/MWh
CEC	Single PV	67.62	No	6%
	Single Wind	77.62	No	6%
	Single Hybrid	69.26	No	6%
	TOU PV	{78.10; 60.67; 76.33; 73.83}	No	5%
	TOU Wind	{76.95; 77.01; 75.87; 79.72}	No	5%
	TOU Hybrid	{74.55; 61.59; 72.01; 71.96}	No	5%
	DAHP	Other fees	Hourly day-ahead	5%
	RTP	Other fees	Hourly day-ahead and balancing	No

Table 4. Main characteristics of the 2019 tariffs.

Analyzing Table 4, it is possible to verify a significant reduction in the prices of the vRES tariffs proposed by the CEC compared with the regulator's tariff. These reductions in tariffs are driven by the lower costs of investing in local renewable generation, but also to the low risk premiums of the CEC compared to competition, and the discounts provided by the regulator to CECs, reflected when CECs are not consuming local energy. Thus, consumers will never select the regulator's tariff, such as simple wind and hybrid tariffs, because better tariffs are available.

Indeed, in the case of solar PV and hybrid tariffs, the mid-peak tariff is the lowest because of the high solar PV production during this period. Commercial consumers may benefit from these reductions by having their highest consumption during mid-peak periods.

The next section presents the selection of tariffs by consumers.

6.2. Consumers' Behavior

In this section, consumers select the tariff that minimizes their costs with electricity for 2019 according to the optimization model of Section 5. Assuming consumers do not have forecast expertise, they select the best tariff using their historical consumption data. Thus, consumers use their consumption in 2018 to compute the best tariff for 2019 according to the options presented in Table 4. Furthermore, consumers also computed their costs with self-consumption and being part of a CEC, which results in the selection of the single PV tariff by 116 consumers, the TOU PV tariff by 1 consumer, and the TOU Hybrid tariff by 195 consumers. The other tariffs were not selected by consumers, such as investing in small-scale self-consumption (rooftop solar PV). Cooperative consumption and coordination of local resources proved to reduce more the costs of consumers than investing in small-scale

self-consumption, because of the lower investments per unit of produced energy and higher efficiency of large-scale renewable technologies, as can be verified by the LCOE values presented in Table 2.

Table 5 presents the final levelized costs of electricity for consumers and their savings regarding the best retail tariffs considering: (i) all proposed tariffs, (ii) the selected tariffs, and (iii) the optimal solution in the case consumers always have the tariff that minimizes their costs. The levelized costs are the average costs per every unit of consumed energy.

Table 5. Final levelized costs of electrical energy for consumers in 2019 considering the different proposed tariffs.

Agents	Tariff	Levelized Price (EUR/MWh)	Savings %
Regulator:	TOU	99.71	-9.01
Retailers:	DAHP	91.47	0.00
	DAMP	93.97	-2.73
	RTP	96.47	-5.47
CEC:	Single PV	67.62	26.07
	Single Wind	77.62	15.14
	Single Hybrid	69.26	24.28
	TOU PV	68.23	25.41
	TOU Wind	77.11	15.70
	TOU Hybrid	67.13	26.61
	DAHP	74.53	18.52
	RTP	71.10	22.27
	Self-consumption	77.52	15.25
	Selected	67.14	26.61
	Optimal	65.12	28.81

Analyzing Table 5, it is possible to conclude that all tariffs proposed by the CEC originate savings regarding the best retail tariff from 15% to 29%. The selected annual tariffs of consumers have a levelized price slightly above 2 EUR/MWh regarding their optimal selection of tariffs every month. This means that the tariffs proposed by the CEC guarantee stable and low prices to members without changing tariffs or suppliers all the time.

Investing in self-consumption is not a good solution for members of the CEC because they will invest more per unit in small-scale PV for lower productivity. However, it is a good option if they have regulated or retail tariffs and do not want to be part of a CEC. Accordingly, to the consumers' selected tariffs, the results of the strategic behavior of the CEC in the wholesale market is presented in the next section.

6.3. Strategic Behavior of the CEC Using Forecasts of Local Net Loads

The selection of vRES tariffs by consumers led the CEC to invest in 745 MW of largescale solar PV and 207 MW of onshore wind using forecast data. Besides the forecast of their members' consumption, the CEC has to forecast the production of its local PV and wind power plants. This investment led the CEC to achieve carbon neutrality and sustainability indexes of 87% and 36%, respectively (the formulation of these indexes is presented in [21]). This means that the CEC generates almost all the energy required by its members (carbon neutrality index), but not necessarily during the period they want to consume it (sustainability index). As a result, it depends on the transport power grid connection and wholesale markets to buy and sell local scarcity and excess energy, respectively. As presented in Section 4, the CEC first forecasts the day-ahead consumption and production of the local members. Its strategy consists of first forecasting local trading to reduce costs with grid usage and then bidding the day-ahead net load into the dayahead market. Against this background, Figure 2 presents the forecast errors of the dayahead forecasts.



Figure 2. Day-ahead forecast errors for demand, net load, and vRES generation.

Analyzing Figure 2 we can conclude that while demand and solar PV forecasts have low errors (notice that PV has no deviations around 50% of the time, since no production is expected), onshore wind have high errors that are reduced in the case of hybrid power forecasts. However, net load forecasts have lower errors, which also supports the incentives provided to CEC to manage local renewable generation. According to Wei et al. [47], the employed forecast methodologies have high forecast accuracies for predicting local demand and net load. Local CECs have the potential to also benefit system operators by reducing local imbalances and contribute to the reliability of the local grid.

Figure 3 presents the monotone curve of the CEC net load in 2019.

Analyzing Figure 3, it is possible to verify that the CEC has more periods with missing energy than excess energy, being its sustainability index equal to 36%. Thus, it will need to buy more energy from wholesale markets than sell it, with its carbon neutrality index being equal to 87%.



Figure 3. Monotone curve of the CEC net load.



Figure 4 presents the levelized costs of the CEC divided per type of cost.



Analyzing Figure 4 it is possible to conclude that the CEC has a levelized cost of 65.16 EUR/MWh, with all required 2019 net load and vRES investments of its members. This means that considering the cost of consumers of 67.14 EUR/MWh (see the "Selected" tariff costs in Table 5), it has a profit of 1.98 EUR/MWh (4.36 million EUR). The CEC may use this profit to guarantee the long-term payment of the vRES investments, but also has backup to compensate consumers when retail tariffs are more attractive than its own tariffs. Thus, CECs may retain members by at least proposing the same tariffs as the competition.

The majority of CEC costs are with annual investing, operation, and maintenance costs with vRES. Trading costs exist because of the small local sustainability index (see Figure 3). However, the CEC earns money in the case of energy excess and pays otherwise. Even so, it buys more energy than sells, with its trading costs being significant (13% of the total costs). Trading, balancing, and other fees costs illustrate the required improvements. Improvements in the forecast accuracy and/or enough flexibility to keep the CEC sustainable and balanced may completely mitigate trading and balancing costs. Furthermore, a fully sustainable CEC may decrease the other fees costs to the medium voltage access costs of 1.38 EUR/MWh. In conclusion, by being part of CECs, consumers may reduce costs by 29%. Furthermore, improvements in forecast accuracy and incentives to demand response programs towards local sustainability may reduce the CEC costs by 22.65 EUR/MWh (35%) to 42.51 EUR/MWh, 54% less than the best retail tariff (see the costs of the DAHP tariff in Table 5). Savings of 54% are aligned with the other studies from the literature (e.g., [29]) Thus, fully sustainable CECs hedge against wholesale price volatility and its future price uncertainty, being their costs based on expected investment and O&M costs with their local resources. Furthermore, CECs can contribute to the local balance and grid reliability since the pricing structure of tariffs incentivize them to be sustainable.

7. Conclusions

This paper studied the behavior of competitive local citizen energy communities participating in wholesale markets. It presented models for optimal investment in local generation assets, tariffs definition, and the strategic bidding behavior of CECs in wholesale markets. Furthermore, it presented a model that simulates the behavior of consumers when selecting tariffs. These models were tested in a computational study using real consumers and market data from the Iberian Market of Electricity (MIBEL).

Using forecasts of market prices and consumption, the CEC designed and proposed eight different tariffs to consumers competing with the regulator and the three most competitive retail tariffs. Those types of tariffs are single, four-rate TOU, DAHP, DAMP, and RTP.

Furthermore, the CEC proposed green tariffs to consumers assigned to local onshore wind, large-scale solar PV, and a hybrid solution with both technologies. Consumers may also invest in their own solar rooftop PV. All consumers select the four-rate solar PV and Hybrid TOU tariffs and the single PV tariffs proposed by the CEC. Their choices led to savings of 27% compared with the best retail tariff. If they always select the best monthly tariff, their savings increase to 29%. Using the strategic bidding and behavior of the CEC in wholesale and retail markets allows the CEC to profit 3% regarding the difference between its members payments and its total costs.

The majority of the costs of the CEC are in its investments in local generation (36%) and their operation and maintenance costs (29%). However, it also has relevant costs on trading (13%), balancing (11%), and other fees (11%) related to grid usage. Trading costs and other fees can be reduced by increasing local sustainability. The local sustainability index is only 34%. Thus, most energy is traded in wholesale markets, even considering that the CEC produces almost all it consumes (carbon neutrality index of 87%). Resource complementarity and demand response shall be used to increase local sustainability and reduce these costs. In addition to these solutions, improving forecast accuracy is also relevant to reduce balancing costs.

Future work aims at improving local resource complementarity and forecast accuracy as well as the inclusion of storage solutions. Furthermore, it also aims at considering the consumers' demand response and dynamic monthly choice of tariffs considering a monthly response of the CEC by proposing new tariffs adapted to the markets and consumers' behaviors.

Author Contributions: Conceptualization, H.A.; Methodology, H.A. and A.C.; Software, H.A.; Validation, A.C., F.L. and A.E.; Formal analysis, H.A., A.C. and F.L.; Investigation, H.A., A.C., F.L. and A.E.; Resources, H.A., A.C., F.L. and A.E.; Data curation, H.A. and A.C.; Writing—original draft, H.A.; Writing—review & editing, H.A, A.C., F.L. and A.E.; Visualization, H.A.; Supervision, A.E.; Project administration, A.E.; Funding acquisition, A.E. All authors have read and agreed to the published version of the manuscript.

Funding: This work received funding from the EU Horizon 2020 research and innovation program under the project TradeRES (grant agreement no. 864276).

Data Availability Statement: Real consumption data for the consumers considered in study are presented in: https://archive.ics.uci.edu/ml/datasets/ElectricityLoadDiagrams20112014#. Iberian market data are available online at: https://mercado.ren.pt/EN/Electr/MarketInfo/MarketResults/OMIE/Pages/default.aspx. Observed and forecast meteorological data are presented in http://www.meteomanz.com/index?l=1. The tariffs proposed in Portugal are available at: (i) regulated TOU, https://www.erse.pt/media/1kifgvjh/estrutura-tarifária-2019-dez2018.pdf, (ii) DAHP, https://www.coopernico.org/files/documents/Coopernico_Tarifario_Indexado.pdf, (iii) DAMP (BTN SPOTDEF) https://luzboa.pt/tarifas/domestico/1, and (iv) RTP (EZU Indexada) https://ezu.pt/tarifas. Data were accessed on 15 March 2024.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

CEC	citizen energy community
СН	Calinski–Harabasz
DAHP	day-ahead hourly pricing
DAM	day-ahead market
DAMP	day-ahead monthly pricing
DR	demand response
DSR	demand-side response
EU	The European Union
GEIC	general economic interest costs
IDM	intraday market
LCOE	levelized cost of energy
MAPE	mean absolute percentage error

MIBEL	Iberian market of electricity
MTS	multivariate time series
NRMSE	normalized root mean squared error
O&M	operation and maintenance
PCA	Principal Components Analysis
PV	photovoltaic
PT	plan template
PMSE	root moon square error
DTD	rool time prising
CEEC	real-time pricing
566	sequential forward feature selection
VRES	variable renewable energy sources
100	time-of-use
Indices	
\mathcal{D}	forecast day
\mathcal{H}	holidays set of days
h	hour
п	year
Ν	life cycle
0	tariff
0	number of tariffs
S	Saturdays set
6	spot market
S	spot market
5	number of spot markets
t	period
T	number of periods
и	voltage level
\mathcal{U}	Sundays set
U	number of voltage levels
υ	technology
V	number of technologies
\mathcal{W}	weekdays set
Parameters	ý
cf_n^v	fixed costs
$C \mathcal{D}^{\mathcal{D}}_{\mathcal{D}}$	variable costs
k	risk premium
nCEC	other fees part of the tariff for CECs
$p_T \hat{r}$	other rees part of the tarm for CLCS
P_{f}	capacity factor
Pn ^v	nominal power
q _{max}	maximum quantity
r	discount rate
Variables	
С	energy cost
С	energy market cost
p_t^{AS}	ancillary services prices
p_t^{DAM}	DAM prices
p_t^{down}	downward imbalance price
p_T^e	energy part of the tariff
n ^f	other fees part of the tariff
PT D ^{prog}	programmed power
i t	evenested anot price
P_{t}	expected spot price
P_t	upward impalance price
9'n	annual generation
<i>q</i> t	expected consumption
$q_{t_{i}}^{cont}$	continuous market quantity
q_t^{dev}	deviated energy
q_t^L	hourly losses
\hat{q}_t^v	expected generation

q̂ ^v ft,pu	expected generation per unit
$\hat{q}_t^{T_h}$	expected hybrid consumption
$\hat{q}_t^{T_v}$	expected consumption per technology
T_t^e	energy retail tariff
w	weight of relative net load

References

- 1. Hunt, S.; Shuttleworth, G. Competition and Choice in Electricity; Wiley: Chichester, UK, 1996.
- 2. Shahidehpour, M.; Yamin, H.; Li, Z. Market Operations in Electric Power Systems; Wiley: Chichester, UK, 2002.
- 3. Kirschen, D.; Strbac, G. Fundamentals of Power System Economics; Wiley: Chichester, UK, 2018.
- Frade, P.; Osório, G.; Santana, J.; Catalão, J. Cooperation Regional coordination in ancillary services: An innovative study for secondary control in the Iberian electrical system. *Int. J. Electr. Power Energy Syst.* 2019, 109, 513–525. [CrossRef]
- Algarvio, H.; Lopes, F.; Sousa, J.; Lagarto, J. Multi-agent electricity markets: Retailer portfolio optimization using Markowitz theory. *Electr. Power Syst. Res.* 2017, 148, 282–294. [CrossRef]
- Algarvio, H.; Lopes, F. Agent-based Retail Competition and Portfolio Optimization in Liberalized Electricity Markets: A Study Involving Real-World Consumers. Int. J. Electr. Power Energy Syst. 2022, 137, 107687. [CrossRef]
- Sekizaki, S.; Nishizaki, I. Decision making of electricity retailer with multiple channels of purchase based on fractile criterion with rational responses of consumers. *Int. J. Electr. Power Energy Syst.* 2019, 105, 877–893. [CrossRef]
- 8. Hong, Q.; Meng, F.; Liu, J.; Bo, R. A bilevel game-theoretic decision-making framework for strategic retailers in both local and wholesale electricity markets. *Appl. Energy* 2023, *330*, 120311. [CrossRef]
- Hampton, H.; Foley, A.; Del Rio, D.F.; Smyth, B.; Laverty, D.; Caulfield, B. Customer engagement strategies in retail electricity markets: A comprehensive and comparative review. *Energy Res. Soc. Sci.* 2022, 90, 102611. [CrossRef]
- Gils, H.; Scholz, Y.; Pregger, T.; de Tena, D.; Heide, D. Integrated modelling of variable renewable energy-based power supply in Europe. *Energy* 2017, 123, 173–188. [CrossRef]
- Lehmann, P.; Creutzig, F.; Ehlers, M.H.; Friedrichsen, N.; Heuson, C.; Hirth, L.; Pietzcker, R. Carbon lock-out: Advancing renewable energy policy in Europe. *Energies* 2012, 5, 323–354. [CrossRef]
- 12. Moro, A.; Lonza, L. Electricity carbon intensity in European Member States: Impacts on GHG emissions of electric vehicles. *Transp. Res. Part D Transp. Environ.* **2018**, *64*, 5–14. [CrossRef]
- 13. Bernath, C.; Deac, G.; Sensfuß, F. Impact of sector coupling on the market value of renewable energies—A model-based scenario analysis. *Appl. Energy* 2021, 281, 115985. [CrossRef]
- 14. Deng, X.; Lv, T. Power system planning with increasing variable renewable energy: A review of optimization models. *J. Clean. Prod.* **2020**, 246, 118962. [CrossRef]
- 15. Nicolini, M.; Tavoni, M. Are renewable energy subsidies effective? Evidence from Europe. *Renew. Sustain. Energy Rev.* 2018, 74, 412–423. [CrossRef]
- 16. Bohland, M.; Schwenen, S. Renewable support and strategic pricing in electricity markets. *Int. J. Ind. Organ.* **2022**, *80*, 102792. [CrossRef]
- 17. Sougkakis, V.; Lymperopoulos, K.; Nikolopoulos, N.; Margaritis, N.; Giourka, P.; Angelakoglou, K. An investigation on the feasibility of near-zero and positive energy communities in the Greek Context. *Smart Cities* **2020**, *3*, 362–384. [CrossRef]
- 18. Tobey, M.; Binder, R.; Yoshida, T.; Yamagata, Y. Urban systems design case study: Tokyo's sumida ward. *Smart Cities* **2019**, *2*, 453–470. [CrossRef]
- Schwidtal, J.M.; Piccini, P.; Troncia, M.; Chitchyan, R.; Montakhabi, M.; Francis, C.; Gorbatcheva, A.; Capper, T.; Mustafa, M.A.; Andoni, M. Emerging business models in local energy markets: A systematic review of peer-to-peer, community self-consumption, and transactive energy models. *Renew. Sustain. Energy Rev.* 2023, 179, 113273 [CrossRef]
- Del Granado, P.C.; Resch, G.; Holz, F.; Welisch, M.; Geipel, J.; Hartner, M.; Forthuber, S.; Sensfuss, F.; Olmos, L.; Bernath, C.; et al. Energy Transition Pathways to a low-carbon Europe in 2050: The degree of cooperation and the level of decentralization. *Econ. Energy Environ. Policy* 2020, *9*, 121–136. [CrossRef]
- 21. Algarvio, H. The Role of Local Citizen Energy Communities in the Road to Carbon-Neutral Power Systems: Outcomes from a Case Study in Portugal. *Smart Cities* 2021, 4, 840–863. [CrossRef]
- 22. Lowitzsch, J.; Kreutzer, K.; George, J.; Croonenbroeck, C.; Breitschopf, B. Development prospects for energy communities in the EU identifying best practice and future opportunities using a morphological approach. *Energy Policy* **2023**, 174, 113414. [CrossRef]
- 23. Gjorgievski, V.Z.; Cundeva, S.; Georghiou, G.E. Social arrangements, technical designs and impacts of energy communities: A review. *Renew. Energy* 2021, *169*, 1138–1156. [CrossRef]
- 24. Neska, E.; Kowalska-Pyzalska, A. Conceptual Design of Energy Market Topologies for Communities and Their Practical Applications in EU: A Comparison of Three Case Studies. *Renew. Sustain. Energy Rev.* **2022**, *169*, 112921. [CrossRef]
- 25. Barnes, J.; Hansen, P.; Kamin, T.; Golob, U.; Musolino, M.; Nicita, A. Energy communities as demand-side innovators? Assessing the potential of European cases to reduce demand and foster flexibility. *Energy Res. Soc. Sci.* **2022**, *93*, 102848. [CrossRef]
- Faia, R.; Soares, J.; Pinto, T.; Lezama, F.; Vale, Z.; Corchado, J.M. Optimal Model for Local Energy Community Scheduling Considering Peer to Peer Electricity Transactions. *IEEE Access* 2021, *9*, 12420–12430. [CrossRef]

- 27. Algarvio, H. Management of Local Citizen Energy Communities and Bilateral Contracting in Multi-Agent Electricity Markets. *Smart Cities* **2021**, *4*, 1437–1453. [CrossRef]
- 28. Klusch, K.; Gerber, A. Dynamic Coalition Formation among Rational Agents. IEEE Intell. Syst. 2002, 17, 42–47 [CrossRef]
- 29. Zepter, J.; Lüth, A.; Del Granado, P.; Egging, R. Prosumer integration in wholesale electricity markets: Synergies of peer-to-peer trade and residential storage. *Energy Build*. 2019, 184, 163–176. [CrossRef]
- European Commission. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee, the Committee of the Regions and the European Investment Bank: Clean Energy for all Europeans (COM/2016/0860 Final). Available online: https://www.eumonitor.eu/9353000/1/j9vvik7m1c3gyxp/vk9u7wdj2bzl (accessed on 15 March 2024).
- 31. European Commission. Regulation (EU) 2019/943 of the European Parliament and of the Council on the Internal Market for Electricity, 5 June 2019. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32019R0943 (accessed on 15 March 2024).
- 32. Gui, E.M.; Macgill, I. Typology of future clean energy communities: An exploratory structure, opportunities, and challenges. *Energy Res. Soc. Sci.* 2018, 35, 94–107 [CrossRef]
- Hoicka, C.E.; Lowitzsch, J.; Brisbois, M.C.; Kumar, A.; Ramirez Camargo, L. Implementing a just renewable energy transition: Policy advice for transposing the new European rules for renewable energy communities. *Energy Policy* 2021, 156, 112435. [CrossRef]
- 34. Kubli, M.; Puranik, S. A typology of business models for energy communities: Current and emerging design options. *Renew. Sustain. Energy Rev.* **2023**, *176*, 113165. [CrossRef]
- 35. Pinto, T.; Morais, H.; Oliveira, P.; Vale, Z.; Praça, I.; Ramos, C. A new approach for multi-agent coalition formation and management in the scope of electricity markets. *Energy* **2011**, *36*, 5004–5015. [CrossRef]
- 36. Algarvio, H. Agent-based model of citizen energy communities used to negotiate bilateral contracts in electricity markets. *Smart Cities* **2022**, *5*, 1039–1053. [CrossRef]
- 37. Yiasoumas, G.; Berbakov, L.; Janev, V.; Asmundo, A.; Olabarrieta, E.; Vinci, A.; Baglietto, G.; Georghiou, G.E. Key Aspects and Challenges in the Implementation of Energy Communities. *Energies* **2023**, *16*, 4703. [CrossRef]
- Ceglia, F.; Marrasso, E.; Pallotta, G.; Roselli, C.; Sasso, M. The State of the Art of Smart Energy Communities: A Systematic Review of Strengths and Limits. *Energies* 2022, 15, 3462. [CrossRef]
- 39. Reis, I.F.; Gonçalves, I.; Lopes, M.A.; Antunes, C.H. Business models for energy communities: A review of key issues and trends. *Renew. Sustain. Energy Rev.* 2021, 144, 111013. [CrossRef]
- 40. Vernay, A.L.; Sebi, C.; Arroyo, F. Energy community business models and their impact on the energy transition: Lessons learnt from France. *Energy Policy* **2023**, *175*, 113473. [CrossRef]
- 41. Lazdins, R.; Mutule, A.; Zalostiba, D. PV Energy Communities–Challenges and Barriers from a Consumer Perspective: A Literature Review. *Energies* 2021, 14, 4873. [CrossRef]
- 42. Algarvio, H. Automated Bilateral Trading of Energy by Alliances in Multi-Agent Electricity Markets. *Electronics* 2023, 12, 2367. [CrossRef]
- Kubli, M.; Canzi, P. Business Strategies for Flexibility Aggregators to Steer Clear of Being "Too Small to Bid". *Renew. Sustain.* Energy Rev. 2021, 143, 110908. [CrossRef]
- Valta, J.; Lummi, K.; Vanhanen, T.; Järventausta, P. Value streams in different energy community types–Review and Implications. In Proceedings of the 2023 19th International Conference on the European Energy Market (EEM), Lappeenranta, Finland, 6–8 June 2023; pp. 1–7.
- 45. Tcholtchev, N.; Schieferdecker, I. Sustainable and reliable information and communication technology for resilient smart cities. *Smart Cities* **2021**, *4*, 156–176. [CrossRef]
- Algarvio, H. Strategic Participation of Active Citizen Energy Communities in Spot Electricity Markets Using Hybrid Forecast Methodologies. Eng 2022, 4, 1–14. [CrossRef]
- 47. Wei, N.; Li, C.; Peng, X.; Zeng, F.; Lu, X. Conventional models and artificial intelligence-based models for energy consumption forecasting: A review. *J. Pet. Sci. Eng.* **2019**, *181*, 106187. [CrossRef]
- Sánchez-Anguix, V.; Aydogan, R.; Baarslag, T.; Jonker, C. Bottom-up approaches to achieve Pareto optimal agreements in group decision making. *Knowl. Inf. Syst.* 2019, *61*, 1019–1046. [CrossRef]
- Heuninckx, S.; te Boveldt, G.; Macharis, C.; Coosemans, T. Stakeholder objectives for joining an energy community: Flemish case studies. *Energy Policy* 2022, 162, 112808. [CrossRef]
- Sousa, J.; Lagarto, J.; Camus, C.; Viveiros, C.; Barata, F.; Silva, P.; Paraíba, O. Renewable energy communities optimal design supported by an optimization model for investment in PV/wind capacity and renewable electricity sharing. *Energy* 2023, 128464. [CrossRef]
- 51. Espe, E.; Potdar, V.; Chang, E. Prosumer Communities and Relationships in Smart Grids: A Literature Review, Evolution and Future Directions. *Energies* **2018**, *11*, 2528. [CrossRef]
- 52. Stadler, M.; Cardoso, G.; Mashayekh, S.; Forget, T.; DeForest, N.; Agarwal, A.; Schönbein, A. Value streams in microgrids: A literature review. *Appl. Energy* 2016, 162, 980–989. [CrossRef]
- Lage, M.; Castro, R.; Manzolini, G.; Casalicchio, V.; Sousa, T. Techno-economic analysis of self-consumption schemes and energy communities in Italy and Portugal. Sol. Energy 2024, 270, 112407. [CrossRef]

- 54. Bistline, J.; Blanford, G.; Brown, M.; Burtraw, D.; Domeshek, M.; Farbes, J.; Fawcett, A.; Hamilton, A.; Jenkins, J.; Jones, R.; et al. Emissions and energy impacts of the Inflation Reduction Act. *Science* **2023**, *380*, 1324–1327. [CrossRef] [PubMed]
- 55. Müller, S.; Welpe, I. Sharing electricity storage at the community level: An empirical analysis of potential business models and barriers. *Energy Policy* **2018**, *118*, 492–503. [CrossRef]
- Anand, A.; Verayiah, R.; Mansor, M.; Hasim, T.; Shukla, A.; Panchal, H.; Sharma, A.; Natrayan, L.; Kumar, A. A comprehensive analysis of small-scale building integrated photovoltaic system for residential buildings: Techno-economic benefits and greenhouse gas mitigation potential. *J. Build. Eng.* 2024, *82*, 108232. [CrossRef]
- 57. Baigorrotegui, G.; Lowitzsch, J. Institutional aspects of consumer (co-) ownership in RE energy communities. In *Energy Transition: Financing Consumer Co-Ownership in Renewables;* Palgrave Macmillan: Cham, Switzerland, 2019; pp. 663–701.
- 58. Singh, N.; Olwagen, R.; Moodley, P. Smart Communities–Embedded Residential Microgrids. In *Domestic Use of Energy*; Eskom Research, Testing & Development: Wellington, South Africa, 2019; pp. 3–7.
- 59. Ledwaba, L.; Hancke, G.; Isaac, S.; Venter, H. Smart microgrid energy market: Evaluating distributed ledger technologies for remote and constrained microgrid deployments. *Electronics* **2021**, *10*, 714. [CrossRef]
- 60. Mansour, K.; Al-Lahham, Y.; Tawil, S.; Kowalczyk, R.; Al-Qerem, A. An Effective Negotiation Strategy for Quantitative and Qualitative Issues in Multi-Agent Systems. *Electronics* **2022**, *11*, 2754. [CrossRef]
- 61. Algarvio, H. Risk-Sharing Contracts and risk management of bilateral contracting in electricity markets. *Int. J. Electr. Power Energy Syst.* **2023**, 144, 108579. [CrossRef]
- 62. Algarvio, H.; Lopes, F.; Couto, A.; Estanqueiro, A. Participation of wind power producers in day-ahead and balancing markets: An overview and a simulation-based study. *Wiley Interdiscip. Rev. Energy Environ.* **2019**, *8*, e343. [CrossRef]
- 63. Brigatto, A.; Fanzeres, B. A soft robust methodology to devise hedging strategies in renewable energy trading based on electricity options. *Electr. Power Syst. Res.* 2022, 207, 107852. [CrossRef]
- Nordström, H.; Söder, L.; Flynn, D.; Matevosyan, J.; Kiviluoma, J.; Holttinen, H.; Vrana, T.K.; van der Welle, A.; Morales-España, G.; Pudjianto, D.; et al. Strategies for Continuous Balancing in Future Power Systems with High Wind and Solar Shares. *Energies* 2023, 16, 5249. [CrossRef]
- 65. ERSE. Tariffs and Prices—Electricity. Available online: https://www.erse.pt/en/activities/market-regulation/tariffs-and-priceselectricity/ (accessed on 15 March 2024)
- ACER. Annual Report on the Results of Monitoring the Internal Electricity and Natural Gas Markets in 2021. Energy Retail and Consumer Protection Volume. October 2022. Available online: https://www.acer.europa.eu/Publications/MMR_2021_Energy_ Retail_Consumer_Protection_Volume.pdf (accessed on 15 March 2024).
- 67. Frade, P.; Pereira, J.; Santana, J.; Catalão, J. Wind balancing costs in a power system with high wind penetration–Evidence from Portugal. *Energy Policy* **2019**, *132*, 702–713. [CrossRef]
- 68. Couto, A.; Estanqueiro, A. Enhancing wind power forecast accuracy using the weather research and forecasting numerical model-based features and artificial neuronal networks. *Renew. Energy* **2022**, 201, 1076–1085. [CrossRef]
- 69. Rodrigues, F.; Trindade, A. Load forecasting through functional clustering and ensemble learning. *Knowl. Inf. Syst.* **2018**, *57*, 229–244. [CrossRef]
- ERSE. Instrumentos para a Participação da Oferta e da Procura na gestão do Sistema Elétrico Nacional. September 2018. Available online: https://www.erse.pt/media/k2kf4glz/instrumentos-para-a-participação-da-oferta-e-da-procura-na-gestãodo-sistema-elétrico-nacional-setembro-2018.pdf (accessed on 15 March 2024).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.